Author's response to reviews

Title: Association of common variants in JAK2 gene with reduced risk of Metabolic Syndrome and related disorders

Authors:

Alberto Penas-Steinhardt Dr. (pufetin@gmail.com)
Mariana L Tellechea Dr. (marianatellechea78@hotmail.com)
Leonardo Gomez-Rosso Dr (mulaq@hotmail.com)
Fernando Brites Dr. (fbrites@hotmail.com)
Gustavo D Frechtel Dr (gfrechtel@ffyb.uba.ar)
Edgardo Poskus Dr (eposkus@ffyb.uba.ar)

Version: 5 Date: 9 December 2011

Author's response to reviews:

BMC Series Editorial Production Team

We are writing to replay your letter regarding the changes to the format of the manuscript, so that it could be published conform to journal style.

Sincerely yours,
Dr Penas-Steinhardt
Corresponding author